Author's response to reviews

Title: Understanding interventions for improving routine immunisation coverage in children in low- and middle-income countries: a systematic review protocol

Authors:

Shingai Machingaidze (missmachi@gmail.com)
Eva Rehfuess (rehfuess@ibe.med.uni-muenchen.de)
Rüdiger von Kries (Ruediger.Kries@med.uni-muenchen.de)
Gregory D Hussey (gregory.hussey@uct.ac.za)
Charles S Wiysonge (charles.wiysonge@uct.ac.za)

Version: 4 Date: 22 October 2013

Author's response to reviews:

Dear Editor,

Thank you for accepting our protocol for publication.

As requested, I have indicated in the Methods section that this protocol has not been registered with PROSPERO both in the Methods section of the abstract, as well as at the end of the Methods section in the main text.

I hope you find these corrections in order.

Kind regards,

Shingai Machingaidze
Research Associate